Point to point reply to reviewers

Reviewer #1:

Scientific Quality: Grade B (Very good)

Language Quality: Grade B (Minor language polishing)

REPLY: we have improved the language with the help of a professional editing service

Conclusion: Accept (General priority)

Specific Comments to Authors:

This FitLiver Cohort Study is an interesting read. However, the authors should increase the sample size in future studies to assure better reliability of results.

REPLY: Thank you for this observation, we will aim to increase sample size in the future.

The spelling of "Newcastle upon Tune, UK)" should be corrected.

REPLY: This have been corrected